Author's response to reviews

Title: Dentist education and labor markets in Mexico. Elements for strategic planning.

Authors:

   Luz Maria LM Gonzalez (luz.gonzalez@insp.mx)
   Gustavo G Nigenda (gnigenda@insp.mx)
   Maria Cecilia MC Gonzalez (cecilia.gonzalez@insp.mx)

Version: 4 Date: 16 January 2012

Author's response to reviews: see over
Editorial Board  
Human Resources for Health Journal  
BioMedCentral  

Dear friends:

Attached to this letter you will find the new version of the document “Dentist education and labor markets in Mexico. Elements for strategic planning”, containing the following modifications:

a) The document was changed from its original case study format and is now in a “Commentary” format.

b) Based on the original format change, now the document has the following sections: a) Background, b) Main Text, c) Results, c) Discussion.

c) The document contain no endnotes so this change was not necessary to make.

We look forward to receive news about the publication of the document.

Best regards

[Signature]

Dr. Gustavo Nigenda  
Co-author